Author’s response to reviews

Title: GatewayNet: A Form of Sequential Rule Mining

Authors:

Phillip C.S.R. Kilgore (pkilgore@lsus.edu)
Nadejda Korneeva (nkorne@lsuhsc.edu)
Thomas C Arnold (TArnol@lsuhsc.edu)
Marjan Trutschl (mtrutsch@lsus.edu)
Urska Cvek (ucvek@lsus.edu)

Version: 2 Date: 13 Nov 2018

Author’s response to reviews:

As requested by the editor, we have attached supplementary files which include a) a copy of the manuscript changes color-highlighted, and b) a document providing point-by-point responses to the editorial and review teams.